Author's response to reviews

Title: Rationale and Design: Telepsychology Service Delivery for Depressed Elderly Veterans

Authors:

Leonard E Egede (egedel@musc.edu)
Christopher B Frueh (frueh@bcm.edu)
Lisa K Richardson (l.richardson@murdoch.edu.au)
Ronald Acierno (acierno@musc.edu)
Patrick D Mauldin (mauldinp@musc.edu)
Rebecca G Knapp (knappr@musc.edu)
Carl Lejuez (clejuez@psyc.umd.edu)

Version: 3 Date: 23 March 2009

Author's response to reviews:

1. The results section has been incorporated into the discussion section as requested.

2. This confirms that the copyright referred to on figure 2 applies to the videophone and not the image. The image is not subject to copyright.